Author's response to reviews

Title: Values and preferences in oral anticoagulation in patients with atrial fibrillation: physicians and patients perspectives: protocol for a two-phase study

Authors:

Pablo Alonso-Coello (palonso@santpau.cat)
Victor Montori (montori.victor@mayo.edu)
Ivan Sola (Isola@santpau.cat)
Holger Schünemann (schuneh@mcmaster.ca)
Devereaux Philipe (philipi@mcmaster.ca)
Charles Cathy (charlesc@mcmaster.ca)
Merce Roura (mrouraol@gencat.cat)
M.Gloria Diaz (mgdiazsanchez@hotmail.com)
Juan Carlos Souto (jsouto@santpau.cat)
Rafael Alonso (ralonsor.gapm10@salud.madrid.org)
Sven Oliver (somarx@gmail.com)
Ignasi Gich (igighs@santpau.es)
Rafael Ruiz (ruizr.bcn.ics@gencat.cat)
Blanca Coll-Vinent (bcvinent@clinic.ub.es)
Ana Isabel Diez (ANAISABEL.DIEZRUlZ@osakidetza.net)
Gordon Guyatt (guyatt@mcmaster.ca)

Version: 4 Date: 20 October 2008

Author's response to reviews: see over
Re: Values and preferences in oral anticoagulation in patients with atrial fibrillation: physicians and patients perspectives.

We have introduced the suggested mention to the ethics committee. We have added a sentence in the last line of the second paragraph of the Design section. It reads as follows:

The study was approved by the ethics committee of the Hospital Sant Pau (Barcelona, Spain).

Yours sincerely,

Pablo Alonso-Coello